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Thursday May 22nd installation

T1041 with LYSO and Ham MCP-PMTs cohabit in the black box. We can leave
our L-bar tests until Tuesday and they can be main user unti then, once ORC
ready.

ORC inspectors (Steve Chappa, Leo Bellantoni et al) come at 10:00 am
tomorrow.

We should have all cables, power supplies, grounding etc ready by then

HV cabling scheme:

T12B-111 cable 111 from patch panel in CR to Cow RR02ch.1 1.4 KV
T22B-112 222 2 12KV
V128113 333 3 11KV
V228114 444 4 11KV
SiPM 2B-121  UUU 5 T2V
HAM12B-101 888 RRO1 1 3KV
HAM2 2B-102 999 2 3KV
PHotekA 28-103 Bertran A 47KV

PHotekB 2B-104  Bertran B  SHV cable special, too short 4.7 KV
18:25 | have to leave now; will be in at 8:30, Anatoly can you come then to wire
up the SiPM before the ORC at 10.:00 77

The ORC was signed off on Friday aftemoon 23rd

Mike

Tuesday May 27th 08:30

Artur and Cal Tech team used the weekend and Monday for LYSO timing
studies.

This morming we take over and install L-bar holder (with one bar at a time) behind
the 2mm x 2mm T1 and in front of one PMT240 ref time. The L-bars will be
pushed against a second PMT240 mounted transversely between bars allowing
sliding to accomodate different LG bar lengths. We will use 2 channels of one
DRS4 box for readout. (The other two can be Artur's transverse lyso bar.)

The trigger has been set up:

T1* T2 and not VA + V2 with beam spil gate.

Citical is positioning of L-bar such that it covers the 2x2 mm T1°T2. Alltriggers
should then have a signal in the Lbar. Parallelism also required.

The bars to be measured are:
QUARTZ [RL]
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T1041 with LYSO and Ham MCP-PMTs cohabit in the black box. We can leave
our L-bar tests until Tuesday and they can be main user until then, once ORC
ready.

ORC inspectors (Steve Chappa, Leo Bellantoni et al) come at 10:00 am
tomorrow.

We should have all cables, power supplies, grounding etc ready by then

HV cabling scheme:

T12B-111 cable 111 from patch panel in CR to Cow RR02ch.1 1.4 KV
T22B-112 222 2 12KV
V128113 333 3 11KV
V228114 444 4 11KV
SiPM2B-121  UUU 5 T2V
HAM12B-101 888 RROT 1 3KV
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